Author's response to reviews

Title: REFLEX: a social-cognitive group treatment to improve insight in schizophrenia study protocol of a multi-center RCT

Authors:

Marieke GHM Pijnenborg (g.h.m.pijnenborg@rug.nl)
Mark M Van der Gaag (m.van.der.gaag@psy.vu.nl)
Claudi CLH Bockting (c.l.h.bockting@rug.nl)
Lisette EM Van der Meer (l.van.der.meer@med.umcg.nl)
Andre A Aleman (a.aleman@med.umcg.nl)

Version: 4 Date: 12 September 2011

Author's response to reviews:

Dear Editor,

I have made the requested major changes:
1) The initials of the authors (if applicable) have been added
2) The sequence of references has been checked and the error in the order of references has been corrected.

I would be most appreciative if you would be so kind to make the minor corrections. Thank you very much in advance.

Yours sincerely, Marieke Pijnenborg